Author’s response to reviews

Title: The Efficacy of Continuous versus Single-injection Femoral Nerve Block in Total Knee Arthroplasty: A Systematic Review and Meta-analysis

Authors:

Hsuan-Hsiao Ma (tyc20192@gmail.com)
Te-Feng Arthur Chou (arthrouchou2@gmail.com)
Shang-Wen Tsai (ray366kimo@gmail.com)
Cheng-Fong Chen (swtsai.vghtpe@gmail.com)
Po-Kuei Wu (drwuvgh@gmail.com)
Wei-Ming Chen (wmchen2499@gmail.com)

Version: 3 Date: 11 Feb 2020

Author’s response to reviews:

Editor Comments:

1. Overlap
We note that the current submission contains some textual overlap with other previously published works, in particular:


This overlap mainly exists in the Methods and Results sections. While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Please be informed that we cannot proceed with handling your manuscript before this issue is resolved, and the sections of text in question have been reformulated.

2. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

=&gt;Thanks. We have revised our manuscripts as your suggestion.